**Author's response to reviews**

**Title:** Neoadjuvant chemotherapy or primary surgery for stage III/IV ovarian cancer: contribution of diagnostic laparoscopy.

**Authors:**
- Jean-Luc Brun (jean-luc.brun@tnn.aphp.fr)
- Roman Rouzier (roman.rouzier@tnn.aphp.fr)
- Frederic Selle (frederic.selle@tnn.aphp.fr)
- Sidney Houry (sidney.houry@tnn.aphp.fr)
- Serge Uzan (serge.uzan@tnn.aphp.fr)
- Emile Daraï (emile.darai@tnn.aphp.fr)

**Version:** 8 **Date:** 30 May 2009

**Author's response to reviews:** see over
There are two minor misprints first line of page 11. Citation of the references [Angioli, Fagotti, Brun] have been replaced by the corresponding numbers from the reference section [12,15].